Author's response to reviews

Title: Interns' knowledge of clinical pharmacology and therapeutics after undergraduate and on-going internship training in Nigeria

Authors:

Kazeem A Oshikoya (med_modhospital@yahoo.com)
Idowu O Senbanjo (senbanjo001@yahoo.com)
Olufemi O Amole (femiamole@yahoo.co.uk)

Version: 6 Date: 22 July 2009

Author's response to reviews:

Figure 1 was initially presented as Figures 1A and B in the same document but only Figure 1A was presented in the core-present version of the manuscript as Figure 1. In this revised copy, the Figures have been saved as to separate documents (Figures 1A and Figure 1B).